```
Mauna Loa Solar Observatory Observer's Log
______
      Wed Apr 9 16:43:33 GMT 2014
Year: 14 Doy: 099
Observer: berkey
WEATHER COMMENT: berkey: Wed Apr 9 16:43:36 GMT 2014
temp 40f, wind 5mph from the S, clear skies inversion layer is capping haze below about 7,000 feet.
 end
Wed Apr 9 16:51:05 GMT 2014: PSPT Start Patrol
GENERAL OBSERVATORY COMMENT BY berkey: Wed Apr 9 16:56:19 GMT 2014
One more last day with observing on the old CoMP PC.
 end
KCOR COMMENT BY berkey: Wed Apr 9 16:57:45 GMT 2014
Connected up the spare fiber Allen pulled down from the spar yesterday. Connected to the R cameralink fiber extender and f
iber ports on the back of the spar. With this PFRemote and CiView can talk to camera 0. Waiting on steves arival for the
T/cameral fiber.
end
Wed Apr 09 17:02:05 GMT 2014 COMP Start Patrol
COMP COMMENT BY berkey: Wed Apr 9 18:11:40 GMT 2014
Testing the copy to mlsoserver script on new comp computer. During this test I did multiple copies of a folder named 2018
5555 with test.txt and test.fts inside. The data is jibberish, but it may show up in LDM and the rest of the pipeline, if
it does plsease either ignore or delete it.
end
Wed Apr 09 18:19:37 GMT 2014 COMP End Patrol
Wed Apr 09 18:19:38 GMT 2014 COMP Start Patrol
Wed Apr 9 19:01:59 GMT 2014: PSPT Start Patrol
PSPT COMMENT BY berkey: Wed Apr 9 19:02:03 GMT 2014
Moved pspt keyboard and mouse control onto the KVM switch.
end
Wed Apr 09 19:10:51 GMT 2014 COMP End Patrol
GENERAL OBSERVATORY COMMENT BY berkey: Wed Apr 9 19:26:06 GMT 2014
Looks like meadowlark controller #2's USB cable was not plugged in this morning, so this mornigns CoMP data may be missing
half of its modulation.
KCOR COMMENT BY berkey: Wed Apr 9 23:19:31 GMT 2014
Kcor is back online. :)
end
COMP COMMENT BY berkey: Thu Apr 10 02:48:59 GMT 2014
CoMP meadowlark controllers are still not talking with labview. This morning we were able to get windows drivers tomorrow
. Hopefully tomorrow we can get new LabView drivers.
____end
```